\begin{table}%T1
\caption{\label{tab1}Datasets contributing to elliptical sample.}
%\centerline
{
\begin{tabular}{lcccc}
\hline \hline
Method$^a$ & Authors$^b$ & Band & Rel. accuracy & Number\\
	&	&	& (rms) &  \\
\hline
FP & G &~$H$ & 0.41 & 41 \\
$L-n$&~$\rm Y_{\mit L}$ &~$B_{J}$ & 0.54 & ~~9 \\
$R-n$&~$\rm Y_{\mit R}$ &~$B_{J}$ & 0.54 & 14 \\
SBF & T &~$I$ & 0.18 & 31 \\
SBF &~$\rm J_{\mit I}$ &~$I$ & 0.07 & ~~6 \\
SBF &~$\rm J_{\mit K}$ &~$K'$ & 0.17 & ~~7 \\
SBF & N &~$I$ & 0.15 & 15 \\
GCLF & K &~$V-I$ & 0.07 & ~~8 \\
\hline 
\end{tabular}}
\smallskip
$^a$ FP = Fundamental Plane;~$L-n,R-n$ = Luminosity / S\'ersic shape
and Radius / S\'ersic shape relations; SBF = Surface Brightness Fluctuations;
GCLF = Globular Cluster Luminosity Function.
$^b$ 
G = \cite{Gavazzi99}; 
T = \cite{Tonry01}; 
$\rm J_{\mit I}$,~$\rm J_{\mit K}$ = \cite{JTL98};
N = \cite{NT00};
K = \cite{KW01}; 
$\rm Y_{\mit L}$,~$\rm Y_{\mit R}$ = \cite{YC95}.
\end{table}